

Explaining the Czech Interbank Market Risk Premium

Geršl, Adam; Lešanovská, Jitka 2013 Dostupný z http://www.nusl.cz/ntk/nusl-155626

Dílo je chráněno podle autorského zákona č. 121/2000 Sb.

Tento dokument byl stažen z Národního úložiště šedé literatury (NUŠL). Datum stažení: 03.05.2024

Další dokumenty můžete najít prostřednictvím vyhledávacího rozhraní nusl.cz .

WORKING PAPER SERIES 1

Adam Geršl, Jitka Lešanovská: Explaining the Czech Interbank Market Risk Premium



WORKING PAPER SERIES

Explaining the Czech Interbank Market Risk Premium

Adam Geršl Jitka Lešanovská

1/2013

CNB WORKING PAPER SERIES

The Working Paper Series of the Czech National Bank (CNB) is intended to disseminate the results of the CNB's research projects as well as the other research activities of both the staff of the CNB and collaborating outside contributors, including invited speakers. The Series aims to present original research contributions relevant to central banks. It is refereed internationally. The referee process is managed by the CNB Research Department. The working papers are circulated to stimulate discussion. The views expressed are those of the authors and do not necessarily reflect the official views of the CNB.

Distributed by the Czech National Bank. Available at http://www.cnb.cz.

Reviewed by: C. Csávás (Magyar Nemzeti Bank) J. Mencía (Bank of Spain) C. Upper (Bank for International Settlements) M. Rusnák (Czech National Bank)

Project Coordinator: Michal Franta

© Czech National Bank, March 2013 Adam Geršl, Jitka Lešanovská

Explaining the Czech Interbank Market Risk Premium

Adam Geršl and Jitka Lešanovská*

Abstract

This paper focuses on the development of the interbank market risk premium in the Czech Republic during the global financial crisis. We explain the significant departure of interbank interest rates from the key monetary policy rate by a combination of different factors, including liquidity risk, counterparty risk, foreign influence, interbank relations, and strategic behavior. The results suggest a relevant role of market factors, and some importance of counterparty risk.

JEL Codes: G19, G21.

Keywords: Counterparty risk, interbank market, liquidity risk, risk premium.

^{*}Adam Geršl, Join Vienna Institute, Czech National Bank, and Charles University, Prague; adam.gersl@gmail.com. Jitka Lešanovská, Czech National Bank and Charles University, Prague;

jitka.lesanovska@cnb.cz.

This research was supported by Czech National Bank Research Project No. A3/2011. The views expressed in this paper are those of the authors and not necessarily those of the Czech National Bank or other institutions with which the authors are affiliated. The authors thank Jan Babecký, Csaba Csávás, Václav Hausenblas, Javier Mencía, Miroslav Plašil, Marek Rusnák, Christian Upper, and Bořek Vašíček for useful comments.

Nontechnical Summary

The paper focuses on identifying individual components of the risk premium in the Czech interbank market which is defined as the departure of 3M PRIBOR rates from the monetary policy 2W repo rate. This spread increased at the beginning of the global crisis, first only a little and then rather abruptly after September 2008. Such an increase in the risk premium negatively influences the transmission of monetary policy to the real economy. Given the foreign ownership of Czech banks and the resulting close links with European banks that were hit by the sub-prime crisis, the uncertainty observed in the international financial markets and the unexpected difficulties of large banks abroad could also have been transmitted to their Czech subsidiaries' and branches' behavior.

We consider counterparty risk and liquidity risk as potentially significant determinants of the interbank risk premium in the Czech Republic. However, we also examine other factors that might significantly influence interbank lending conditions in the Czech Republic, such as market conditions in relevant domestic as well as international markets, interbank lending relations of banks, and possible strategic behavior. Following a review of the literature on interbank markets, an analysis of the specific features of the Czech interbank market, and a description of interviews conducted with reference banks' representatives, we define five hypotheses relating to the individual components of the risk premium and test them using the Bayesian Model Averaging (BMA) method.

Daily data on reference banks' quotes for PRIBOR fixing as well as other daily and monthly data on market developments and banks' balance-sheet positions are used. We aim at explaining the banks' behavior when providing quotes for 3M PRIBOR fixing and define the interbank risk premium as the departure of 3M PRIBOR rates from the monetary policy repo rate. The explanatory variables include liquidity and solvency indicators of quoting (reference) banks; the averages and standard deviations of these variables across potential interbank counterparties to capture counterparty risk; market variables such as bond market liquidity, the euro interbank risk premium, and credit default swap spreads; indicators capture possible strategic behavior, for example the ratio of loans linked to PRIBOR to total assets and the sensitivity of the interest rate derivatives portfolio to PRIBOR changes. We perform our analysis both for the whole period 2007–2012 and separately for the crisis period (2008–2009) and the recession period (2009–2012).

The results using daily data show significant persistence of the risk premium and a relevant role of market variables such as bond market liquidity and the euro area interbank risk premium. We find some role of counterparty risk, as measured by the average ratio of non-performing loans to total client loans of potential interbank counterparties and the dispersion of the credit default swap spreads of parent companies of potential interbank counterparties. The latter variable indicates possible cross-border contagion effects via foreign ownership on the Czech interbank market risk premium. We do not find robust evidence for strategic behavior of banks when quoting PRIBOR rates.

1. Introduction

The interbank markets play an important role in many aspects in almost all economies. They serve as a source of funding and a venue for liquidity management of banks. Moreover, interbank markets play a key role in the monetary policy transmission mechanism and represent one of the main channels through which the official interest rates set by central banks are transmitted further to commercial bank rates and to the real economy. Finally, interbank market rates reflect the actual situation of the demand for and supply of liquid funds in the market and play an important role as reference rates for a number of financial products – be they selected derivatives, bank loans or deposits.

During the global financial crisis which started in the US sub-prime mortgage segment in 2007, interbank markets worldwide experienced an increase in volatility and decrease in liquidity. While interbank market rates usually closely mirror the monetary policy rate in normal times, a quite strong departure of interbank rates from the key central bank rate has been observed since the onset of the global financial crisis. This increase in risk premium and a parallel significant decrease in interbank market liquidity also happened in the Czech interbank market despite the fact that Czech banks were not directly hit by losses from subprime-related structured products.

The objective of this paper is to identify individual components of the interbank risk premium in the Czech interbank market during the global financial crisis period. We consider credit and liquidity risk as potentially significant determinants of the interbank risk premium in the Czech Republic. However, we also examine other factors that might significantly influence interbank lending conditions in the Czech Republic. Specifically, we examine the role of other bank-specific characteristics (e.g., market share, interbank lending relations, portfolio composition of the bank) and the situation in the relevant domestic as well as international markets.

The paper is structured as follows. The following section 2 is devoted to a review of relevant literature, while section 3 describes the Czech interbank market. Section 4 introduces the hypotheses that are tested. Section 5 describes the data used for the analysis and explains the methodology. Section 6 discusses the results and section 7 concludes.

2. Literature Review

A number of theoretical and empirical papers examine interbank markets and try to explain the interbank market freeze during the global financial crisis. Heider et al. (2009) show that the unsecured interbank market freeze stems mainly from counterparty risk and the asymmetric information problem, where an interbank market breakdown occurs when average counterparty risk and the dispersion of counterparty riskiness are high. Similarly, Freixas and Jorge (2008) find that asymmetric information might lead to credit rationing in the interbank market.¹

¹ Credit rationing is the situation where demand for credit exceeds supply and borrowers do not receive the full amount of credit they require or some of the borrowers are completely turned down (Afonso and Aubyn, 1998).

According to Gai et al. (2011) an interbank market freeze can occur not only due to the counterparty credit risk, but also due to liquidity hoarding, possibly with the precautionary motive.² The precautionary motive becomes more important as the funding risk of the "creditor" bank increases. In their model, hoarding behavior by one bank leads to liquidity hoarding by other banks, where the probability and scope of such a domino effect is dependent on the interbank market structure. Although precautionary hoarding can introduce reputational costs for the hoarding bank, the authors assume such costs to be lower than the costs resulting from fire sales of banking assets.

Caballero and Simsek (2009) emphasize the role of the complexity of the interbank market structure and the cost of information gathering in a potential interbank market freeze. In normal times, it might be sufficient for a bank to assess the riskiness of its direct counterparties only. However, when financial distress occurs, the bank should become better informed not only about its direct trading partners, but also about the trading partners of its trading partners, and so on. Information gathering becomes impossible during the crisis due to the complex interbank structure, so the bank decides to withdraw its loan commitments from its counterparties and starts hoarding liquidity. This can lead to a credit crunch.

Moreover, Acharya et al. (2010a) show that if market power is concentrated in highly liquid banks, borrowing becomes too costly for banks that need liquidity even when there is no shortage of liquidity on the aggregate level. Weaker banks are then left with the option of either using expensive interbank borrowing or selling their assets. Banks with a liquidity surplus will then be able to purchase those assets at a price well below their fundamental value (predatory behavior). Acharya et al. (2010a) provide a rationale for the role of a central bank as a lender of last resort, enabling small banks to borrow from the central bank and to avoid being under pressure from banks with predatory intentions.

Similarly to the theoretical literature, empirical papers on the interbank market assume credit and liquidity risk factors to be the main drivers of the interbank risk premium during the global financial crisis. The pioneering testing of this hypothesis was undertaken by Taylor and Williams (2008) and Michaud and Upper (2008). Both studies use the London Interbank Offer Rate (LIBOR) over the overnight indexed swap³ (OIS) spread.⁴ However, the results of these studies differ. Michaud and Upper (2008) suggest that the increase in the interbank premium was driven mainly by liquidity factors, specifically by concerns about the banks' own funding liquidity position, and that the credit

 $^{^2}$ Since the functioning of the interbank market might be disrupted in the crisis, banks might be uncertain about whether they will be able to obtain sufficient liquidity in the market in future if they are hit by a liquidity shock (i.e., significant withdrawals by depositors). Hence, banks might become reluctant to lend to other banks and instead keep sufficient liquidity in their balance sheets. Moreover, such behavior might be more apparent among banks with weaker balance sheets, since these banks might be facing even higher funding risk. Acharya and Merrouche (2010) found that banks with higher losses were more prone to liquidity hoarding during the crisis.

 $^{^{3}}$ An overnight indexed swap is an interest rate swap where the floating leg of the contract is linked to a particular index of overnight interest rates (for example EONIA in the euro area or CZEONIA in the Czech Republic). Given its construction – the counterparties exchange only the difference between the fixed and floating rate times the nominal value of the contract – it should be (almost) free of counterparty credit risk and liquidity risk premium and capture expectations of future short-term interest rates well.

⁴ The spread between LIBOR and OIS should capture only the risk premium in the interbank market, as both the LIBOR rate and the OIS rate incorporate expectations of future short-term interest rates, which are mainly determined by central bank monetary policy. Monetary policy expectations are therefore cancelled out by deducting the OIS rate from the LIBOR rate.

risk measured by credit default swap (CDS) spreads⁵ is transmitted to LIBOR rates over a longer period of time. Taylor and Williams (2008), on the other hand, focus solely on the US interbank market and find a strongly significant effect of counterparty credit risk on the interbank premium, but a low impact of liquidity risk. Liquidity risk seemed to be of minor importance, since the impact of the Term Auction Facility of the Federal Reserve System on the interbank premium turned out to be insignificant.

Nobili (2009) and Kamps (2009) examine the risk premium in the European interbank market (EURIBOR-OIS spreads) during the global financial crisis. They find that both credit risk and liquidity risk were sources of the excessive EURIBOR-OIS spreads, and Nobili (2009) shows that their relative importance varied over time. Liquidity risk was more apparent at the beginning of the crisis and its weight significantly weakened in the last quarter of 2008. This decrease in liquidity risk is attributed to the effectiveness of the ECB's unconventional monetary policy.

Angelini et al. (2009) focus on interbank market rates of longer maturities (beyond one week) before and during the global crisis. Unlike the previous studies, which looked at aggregate reference rates, their analysis is conducted using micro data on real euro-denominated transactions executed by Italian and other European banks further merged with bank-specific characteristics. Their results show that in the pre-crisis period, borrowers' capital and liquidity position played no significant role, while lenders' capital and liquidity position mattered, i.e., more liquid and less risky lenders provided their funds at lower rates. Lenders' behavior has changed since the onset of the crisis, with highly liquid banks charging higher interest rates. They conclude that this might be in line with the predatory behavior suggested by Acharya et al. (2010a) or it may reflect idiosyncratic risk aversion of liquidityrich banks. Moreover, the effect of borrower capitalization became significant during the crisis.

Acharya and Merrouche (2010), analyzing the sterling secured and unsecured interbank markets, find precautionary hoarding due to increased funding liquidity risk to be the main factor contributing to the elevated interbank market spreads in the UK, in line with the theoretical explanation by Gai et al. (2011). However, their analysis only includes the crisis period between the start of the crisis in August 2007 and June 2008, whereas the role of counterparty credit risk might have become more crucial in the post-Lehman period. Moreover, the stress on the secured market might also have been influenced by a decrease in the debt capacity of assets⁶ used in repo transactions. Acharya et al. (2010b) show that the debt capacity of an asset can decrease quickly when the rollover frequency of debt is sufficiently high, even though the quality of the asset is respectable.

3. The Czech Interbank Market and the Role of PRIBOR

3.1 Characteristics of the Czech Interbank Market

The interbank market is used by commercial banks to exchange short-term liquid funds among themselves at a pre-specified price. In a wider sense, the interbank market also includes the transactions of commercial banks with the central bank. The interbank market is of key relevance for

⁵ A credit default swap (CDS) is a financial derivative that allows the security holder to insure against debtor default. The CDS price paid by the buyer of the insurance, called the CDS spread, should therefore be a good indicator of counterparty credit risk.

⁶ The debt capacity of an asset represents the size of borrowing that can be obtained using the asset as collateral.

the monetary transmission mechanism, since monetary policy rates are expected to be transmitted to interbank market rates and on to the real economy (the interest rate channel), possibly also influencing the amount of credit to the private sector due to balance-sheet effects (the credit channel). Moreover, interbank rates – especially "reference" rates such as LIBOR, EURIBOR, and (in the Czech Republic) PRIBOR – often serve as base rates for pricing different financial products, both retail (client loans and deposits) and wholesale (derivatives such as interest rate swaps). While the interbank market serves as an important source of funding for banks in a number of countries (e.g., the UK, Italy, and Sweden) banks in the Czech Republic use interbank lending rather to balance their short-term liquidity needs.

The size of the Czech banking sector has risen significantly over the last decade. The total assets of Czech banks have doubled since 2000 and currently amount to roughly 120% of GDP, while the number of banks has increased only marginally (from 40 in 2000 to 44 in 2012). The Czech banking sector consisted of 18 banks, 21 foreign bank branches and 5 special-purpose building societies in 2012. The development of the Czech banking sector has been influenced by large-scale privatizations of large, previously state-owned banks in the early 2000s as well as by subsequent mergers and acquisitions.⁷ A distinct feature of the Czech banking sector is that the majority of banks are foreign owned.⁸

The Czech banking system exhibits favorable values of key indicators such as profitability and solvency (capital adequacy exceeding 15%), a relatively low level of non-performing loans, and, unlike Hungary or Poland, no household FX loans. Moreover, there is an overhang of deposits over loans, leading to a structural excess of liquidity in the sector. This contributes to a favorable liquidity position of most banks and no dependence on foreign funding (CNB, 2012). This holds not only for the pre-crisis period, which was marked by respectable credit growth, but also for the peak of the 2008–2009 crisis. Due to the strong position of the banking sector there have been no bank failures or bank runs since the global financial crisis started and thus there has been no need for public intervention (Frait et al., 2011).

The structural excess of liquidity in the domestic banking sector is regularly absorbed by the CNB via its repo operations, unlike in the euro area, where the ECB injects liquidity into the banking sector.⁹ In October 2008, following a liquidity crisis in the Czech government bond market, the CNB launched a temporary liquidity-providing facility with 2W and 3M maturities, in which Czech government bonds can be used as collateral.¹⁰ This facility proved to be of great importance, since around 15% of banking assets consist of government bonds (CNB, 2012). It helped to calm the government bond market (as holders were reassured that they could always exchange it for liquidity using the CNB facility). Given the continuing stress in the international financial markets in the period 2008–2012, it is still in place for prudential reasons, although only for 2W maturity. Nevertheless, there is very often no interest in the auctions; the facility is used rather sporadically.

⁷ The development of the Czech banking sector is discussed, for example, in Bárta and Singer (2006).

⁸ Almost 97% of the banking sector's balance-sheet assets are controlled (directly or indirectly) by foreign owners (CNB, 2012).

 $^{^{9}}$ In total, roughly 10% of aggregated banking sector assets are deposited with the CNB as a result of these repo operations.

¹⁰ See Geršl and Komárková (2009).

An important feature of the Czech interbank market is its relatively low market liquidity not only in the crisis period, but also in the pre-crisis period. Average trading volumes of unsecured Czech koruna deposits amount to around CZK 60 billion (more than half of which conducted with non-resident banks) and are heavily concentrated at the shorter end of the maturity curve (O/N trading usually represents around 80% of daily trading volumes).¹¹ Interbank trading on a secured basis has increased since the onset of the global financial crisis, but there is still much less activity in comparison to unsecured lending. Finally, Czech banks are quite active in trading interest rate derivatives linked to interbank market rates, especially forward rate agreements (FRA) and interest rate swaps (IRS).

3.2 Specifics of PRIBOR

Unlike in the case of EURIBOR and LIBOR, reference interbank market rates are quoted for both the bid (PRIBID, Prague Interbank Bid Rate) and offer (PRIBOR, Prague Interbank Offered Rate) sides of the contract in the Czech Republic. The benchmark PRIBOR and PRIBID are calculated from individual quotes submitted by so-called reference banks. Reference banks are supposed to be significant participants in the interbank market for deposits and products derived from interest rates, even under rapidly changing market conditions (CNB, 2006).

The official definition of PRIBOR differs from the definition of EURIBOR and LIBOR. PRIBOR is defined as the reference interest rate on the interbank deposit market calculated (fixed) by the calculation agent for the Czech Forex Club from the quotations of reference banks for the sale of deposits (CNB, 2006).¹² The "quotation" is the price at which a reference bank is willing to sell an interbank deposit to another reference bank (CNB, 2006). EURIBOR is the rate at which euro interbank term deposits are offered by one prime bank to another prime bank within the EMU zone, and is published at 11.00 a.m. (CET) for spot value (T+2) (European Banking Federation). In contrast, LIBOR is the rate at which an individual contributor panel bank could borrow funds, were it to do so by asking for and then accepting interbank offers in reasonable market size, just prior to 11.00 a.m. London time (British Bankers' Association).

Since the definitions of LIBOR, EURIBOR, and PRIBOR differ, they might not be fully comparable and might be associated with different difficulties. Quotes for the benchmark LIBOR might be biased downwards, since LIBOR banks should indicate via their quotes how costly it would be to secure funding for themselves. By providing lower quotes, LIBOR banks might prevent negative signaling about their possibly tight funding risk position (Kamps, 2009). The EURIBOR rate might be influenced by the unclear definition of the term "prime bank," as the notion of prime bank might differ across the banks providing quotes for EURIBOR and also across time (Taboga, 2013). In the case of PRIBOR, both the lender and the borrower (counterparty bank) are exactly defined, as both of them are reference banks. However, the Czech banking sector is rather small by international comparison and the dozen or so reference banks providing quotes for PRIBOR differ significantly from each other in various respects, such as the size of the bank, its legal form (subsidiaries versus branches of foreign banks), and riskiness. Hence, it might be challenging for a representative of a

¹¹ See the results of the regular money market turnover survey on the CNB website: http://www.cnb.cz/en/financial_markets/money_market/mm_turnover/.

¹² The reference bank submits data for the calculation of the PRIBID and PRIBOR reference rates every business day between 10.30 a.m. and 10.45 a.m. local time. The calculation of PRIBID and PRIBOR takes place at 11.00 a.m. local time and the PRIBOR and PRIBID rates are published immediately after the calculation is completed.

reference bank to provide an appropriate quote for such a heterogeneous group of potential counterparties.¹³

PRIBOR and PRIBID reference rates are calculated as the mathematical arithmetic mean rounded to two decimal places for overnight (O/N) maturity and for maturities of 1 and 2 weeks and 1, 2, 3, 6, 9, and 12 months. Since there are 11 reference banks in the PRIBOR and PRIBID panel as of June 2012, the two highest and two lowest quotes are removed from the calculations in order to prevent deliberate manipulation of the PRIBOR index.¹⁴ The resulting PRIBOR and PRIBID rates are not binding, i.e., banks only indicate what would be the approximate price for lending to or depositing the money of other reference banks. The real price might differ from the quoted price in PRIBOR and PRIBID.¹⁵

Figure 1: Share of Interbank Exposures in Balance Sheet of Reference Banks (%)



Note: Exposures to resident banks in CZK. *Source:* CNB

Figure 2: Number of Reference Banks Over Time





The 11 reference banks account for 69% of the total assets of the Czech banking sector and for 51% of total domestic interbank lending in CZK as of June 2012. On average, however, the interbank exposures of the reference banks as a proportion of their assets are rather small (Figure 1). The panel of reference banks comprises all four large banks, five branches of foreign banks, and two small and medium-sized banks. All reference banks are under foreign control.

However, the number of reference banks has varied between 11 and 13 over the last decade (Figure 2). Moreover, there have been some mergers and acquisitions (M&As) during the period, influencing not only the size and market shares of the banks concerned, but also their ownership structure and

¹³ There are other additional differences between LIBOR, EURIBOR, and PRIBOR fixing, e.g. differences in the number of panel banks or the number of quotes that are excluded from the calculation of the benchmark rates.

¹⁴ When 11 or more banks provide quotes, the two highest and two lowest quotes are removed from the calculations. When the number of quotes is between 6 and 10, the highest quote and the lowest quote are excluded. When the number of quotes is 4 or 5, all the quotes are included in the calculation sample. When the number of quotes is less than 4, the interest rate for the relevant maturity is not fixed.

¹⁵ Unfortunately, data on whether the interest rate on real transactions is close to the quoted PRIBOR rates are not available.

related corporate governance structure (domestically owned bank vs. subsidiary of foreign bank vs. branch of foreign bank).¹⁶

Out of the nine maturity buckets for both PRIBOR and PRIBID (O/N, 1W, 2W, 1M, 2M, 3M, 6M, 9M, and 12M), the 1M, 3M, and 6M PRIBORs are the most followed reference interbank rates, since they represent the base rate for pricing financial products such as interest rate derivatives and certain deposits and loans, mainly to non-financial corporations. In comparative analyses and models, the 3M PRIBOR plays an important benchmark role. Also, the official forecasting model of the CNB (the DSGE-type "g3" model) includes the 3M PRIBOR as a market proxy for the monetary policy. In this paper, we focus mainly on the 3M PRIBOR, acknowledging the fact that the other PRIBORs are correlated with the 3M rate to a large extent.

3.3 Development of the Risk Premium in the Czech Interbank Market

Prior to the global crisis, the situation in the domestic interbank market was stable and the 3M PRIBOR closely mirrored the key monetary policy rate (the CNB's 2-week repo rate), resulting in a very tight spread between these two rates, with its occasional variation mainly reflecting short-term (3-month) expectations of changes in the monetary policy rate (Figure 3). However, the 3M PRIBOR–2W repo spread increased at the beginning of the global crisis, first only a little and then rather abruptly after September 2008. Given the foreign ownership of Czech banks and the resulting close links with European banks that were hit by the sub-prime crisis, the uncertainty observed in the international financial markets and the unexpected difficulties of large banks abroad could also have been transmitted to their Czech subsidiaries' and branches' behavior.

¹⁶ First, BAWAG Bank CZ merged with BAWAG International Bank CZ (previously Dresdner Bank) on March 31, 2005. BAWAG Bank CZ became a reference bank instead of BAWAG International Bank CZ. Moreover, the change of ownership of BAWAG Bank CZ on September 1, 2008 was also connected with the rebranding of the bank to LBBW Bank CZ. LBBW Bank CZ continued to serve as a reference bank. Second, the merger of HVB Bank Czech Republic and Živnostenská banka reduced the number of reference banks from 13 to 12, since both of them had been providing quotes in PRIBOR/PRIBID fixing until then. The result of the merger on November 5, 2007 was UniCredit Bank Czech Republic, the fourth biggest bank in the Czech banking sector, which continued to be a reference bank in PRIBOR/PRIBID fixing. Third, Raiffeisenbank started providing quotes for PRIBOR/PRIBID fixing at the beginning of March 2007. This temporarily increased the number of reference banks to 13. On July 7, 2008, Raiffeisenbank and eBanka completed their merger and Raiffeisenbank continued providing quotes in PRIBOR/PRIBID fixing. Fourth, Crédit Agricole (in that time CALYON BANK, previously Crédit Lyonnais) and Citibank Europe changed their governance structures from a subsidiary to a branch of a foreign bank at the end of 2005 and 2007/2008, respectively. Both remained reference banks. Fifthly, ABN AMRO Bank N.V. changed its name to The Royal Bank of Scotland N.V. in 2010.

Figure 3: Risk Premium in Czech and Euro Interbank Markets (bp)



Figure 4: Coefficient of Cross-sectional Variation of 3M PRIBOR Quotes and Bid-ask Spread (%, bp)



Source: CNB, Thomson Reuters Datastream

Source: CNB, Thomson Reuters, authors' calculation

In parallel to the increase in the 3M PRIBOR–2W repo spread there was a significant rise in the bidask spread in the interbank market, connected with higher cross-sectional variation of the individual quotes from the end of September 2008 onwards as measured by the standard deviation of the individual quotes divided by the mean (Figure 4). Until then, the market practice had been to keep the spread between the quotes submitted for PRIBOR and PRIBID at 10 bp, limiting arbitrage opportunities in the interbank market. This practice, however, stopped in September 2008, as some reference banks – reflecting the conditions then prevailing in international financial markets – increased the quoted PRIBOR independently of their quoted PRIBID. As a result, the PRIBOR-PRIBID spread increased to around 40 bp, which also contributed to higher dispersion of PRIBOR and PRIBID quotes.

In order to gain more information on the market practice as regards PRIBOR quoting, we conducted several interviews with reference bank representatives. The main messages can be summarized as follows. First, there was a clear structural change in PRIBOR quoting in September/October 2008, when banks stopped quoting a PRIBOR-PRIBID spread of 10 bp and started to take more account of various risks related to interbank market lending (counterparty, liquidity, etc.) as well the general uncertainty in markets and the economy. The preceding period (say 2002-early 2007) had been rather exceptional from today's perspective, as there had been a high persistence in quoting, the main driver had been monetary policy expectations, and the role of counterparty and other risks had been limited. Second, the liquidity in the Czech money market has always been and remains rather low (with the exception of the O/N segment), so the quotes should be understood only as reference or benchmark rates. Actual interest rates in trading with other banks could differ from the reference rate. Third, charging different rates for different counterparties – although also used to some degree – was not the main strategy. Rather, banks increased or reduced their credit lines/limits for particular counterparties and/or the maturity of interbank loans to reflect counterparty risk. In addition, the risk management strategy was often influenced or even set by the parent bank. Moreover, when assessing counterparty risk, banks looked at the situation of parent banks of possible domestic counterparties, including their CDS spreads. Finally, when providing quotes to PRIBOR, banks took significant account of market developments such as euro area spreads and other market variables.

4. Hypotheses

The literature overview provides a number of potential hypotheses that could be tested in the Czech context. Generally, the increase in the interbank risk premium is seen as a consequence of liquidity hoarding by banks reluctant to lend in the interbank market during the global crisis. However, there are various possible sources of such hoarding behavior. Heider et al. (2009), Freixas and Jorge (2008), and Taylor and Williams (2008) suggest that the source of liquidity hoarding is counterparty credit risk, while Gai et al. (2011), Acharya and Merrouche (2010), and Michaud and Upper (2008) explain liquidity hoarding behavior by concerns about the banks' own liquidity position.

Precautionary hoarding due to liquidity effects is influenced not only by the current liquidity position of a bank, but also by expectations about the future development of the market situation and banks' solvency position. For example, expectations of increasing credit risk losses on the loan portfolio might reduce liquidity inflows (interest income) and increase liquidity risk in terms of a possible bank run should the bank come closer to insolvency.

Further, the complex and highly interconnected structure of the interbank market might influence banks' behavior in the interbank market. The complexity of interbank relationships might increase the interbank risk premium in crisis times (Caballero and Simsek, 2009). However, the market complexity hypothesis is difficult to test, since banks' representatives quoting PRIBOR do not directly observe the complexity of the market over time. On the other hand, long-term lending relationships between banks (relationship lending) and high connectivity of the reference banks with other reference banks (reference banks are very well known counterparties) might decrease the interbank risk premium.

Besides the above-mentioned factors discussed in the theoretical and empirical literature, specific features of the Czech banking sector are also taken into account. First, conditions abroad might be relevant for PRIBOR quoting by domestic reference banks due to their foreign ownership and specific corporate governance structure. A worsening of the liquidity and solvency position of a parent bank could be transmitted to increased PRIBOR quotes provided by its Czech subsidiary, since the parent bank usually needs to satisfy some liquidity and solvency standards on the consolidated banking group level, so the parent bank might aim to influence the behavior of its subsidiary on the Czech interbank market.¹⁷ Moreover, the counterparty risk of domestic banks might not be viewed by the quoting reference bank as completely separated from the riskiness of the counterparty's parent bank. Second, given the high exposure of domestic banks to Czech government bonds, which account for an important part of their liquid assets, market conditions in this particular asset market are covered by our data, as they directly influence banks' balance-sheet positions. Finally, we also reflect individual banks' business models in the data construction.

More recently, the LIBOR manipulation scandal revealed that the quoting behavior of individual reference banks might be driven by strategic considerations and attempts to manipulate the final

¹⁷ Eventually, the risk of subsidiaries could also be transmitted to parent banks, influencing the riskiness of the whole banking group. However, the Czech subsidiaries of Western European banks did not increase the risk of the group compared to the losses of the parent bank or other subsidiaries. On the contrary, even in the crisis period the highly profitable Czech banks helped stabilize the overall banking group position.

benchmark rate.¹⁸ In the Czech context, such strategic behavior could emerge if there were benefits in doing so in terms of profitability considerations. Banks in the Czech interbank market could have had an incentive to report higher rates in order to increase profits on financial products held in their balance sheets and linked to PRIBOR, such as PRIBOR-linked loans to non-financial corporations or interest rate derivatives such as FRA and IRS.¹⁹ However, while the potential incentive to manipulate PRIBOR upwards to generate income on PRIBOR-linked loans exists across all banks, the direction of potential manipulation of PRIBOR to influence the value of derivatives held and linked to PRIBOR can differ across banks, as net exposures and their sensitivity to changes in PRIBOR also differ. Moreover, predatory behavior could also have played a role given the good liquidity position of some reference banks and the strong concentration of the Czech banking sector, in which the four biggest banks have dominant positions.

In the empirical investigation, we test the following hypotheses:

H1: Higher funding liquidity risk of the reference banks, as measured by liquidity and – due to their impact on possible funding liquidity risk – solvency risk indicators, including indicators of the foreign parent bank, increases the risk premium in the domestic interbank market.

H2: Counterparty risk, as measured by indicators of potential counterparties and - allowing for contagious effects - of their foreign parent banks, increases the risk premium in the domestic interbank market.

H3: Market variables, especially higher uncertainty, lower liquidity, and higher volatility in other domestic and foreign markets, contribute to the risk premium in the domestic interbank market.

H4: Higher connectivity of reference banks among themselves, together with long-term interbank lending relationships (relationship lending), increases the knowledge of reference banks about each other, leading to a decrease in the interbank risk premium.

H5: Some strategic behavior by domestic reference banks emerged during the crisis period.

¹⁸ See UK Treasury Wheatley Review of LIBOR: Initial Discussion Paper, available at http://www.hm-treasury.gov.uk/d/condoc_wheatley_review.pdf.

¹⁹ Mortgages and consumer loans are traditionally provided with longer interest rate fixations (3–5 years being the most common period over which the interest rate is fixed in the case of mortgages).

5. Empirical Approach

5.1 Data

Data for the empirical analysis were taken from various sources. Firstly, a full dataset comprising PRIBOR and PRIBID quotes by individual reference banks was obtained from Thomson Reuters. Secondly, reference banks' bank-level data were extracted from CNB supervisory reporting and the CNB's credit register. Thirdly, market data were obtained from Bloomberg and Thomson Reuters Datastream. The bond market liquidity index is internally calculated by the CNB using the methodology described in Geršl and Komárková (2009).

All variables used in the empirical analysis are summarized in Appendix A in the structure defined by the hypotheses. We construct a number of ratios and indicators following the related literature as well as our judgment as to what could best capture the determinants of the quoting behavior of reference banks. The reference-bank-specific data, originally at monthly and quarterly frequency, were linearly interpolated in order to obtain a daily data set, in contrast to the counterparty banks' characteristics, where constant observation was applied, reflecting the information available to the representatives of the reference banks at the time of PRIBOR quoting. Alternatively, to obtain the monthly dataset used for the robustness check of the results based on daily frequency, monthly averages of the daily variables were created.

The interbank risk premium in the Czech interbank market is represented by the 3M PRIBOR less the 2W repo. We use this spread to capture the domestic interbank risk premium – despite the fact that it includes a term premium and expectations about the monetary policy rate over the next 3 months – for three main reasons. Firstly, the alternatives (such as the PRIBOR–OIS spread²⁰) are not reliable due to the low liquidity of the CZK OIS market. Secondly, the 3M PRIBOR–2W repo spread is often publicly discussed, since monetary policy analysts compare the repo rate and the benchmark 3M PRIBOR rate. Thirdly, the official CNB "g3" forecasting model includes this premium, so the results of this analysis can help complement the forecasts based on the g3 model. Moreover, we control for expected monetary policy changes using the 1x4 FRA rate, which captures expectations about future monetary policy in the immediate future. We also include dummies for individual banks to capture bank-specific effects.

The first group of explanatory variables comprises funding liquidity and solvency indicators which indirectly influence *funding liquidity risk*, such as bank run risk if a bank's solvency position deteriorates, to test the H1. Liquidity indicators are represented by traditionally used indicators such as the ratio of liquid assets²¹ to total assets and the loan-to-deposit ratio. We cannot use the traditional solvency indicators, such as the regulatory capital adequacy ratio or the capital to assets ratio, as a number of reference banks are foreign bank branches without regulatory capital. Thus, we use variables that are related to or influence solvency, i.e., credit risk indicators of banks' portfolios (the

²⁰ The difference between the interbank market rate (e.g., EURIBOR) and the OIS of the corresponding maturity and currency is commonly used in the literature as an approximation of the interbank risk premium. This expression of the interbank risk premium has the advantage that expectations about monetary policy are effectively removed from the spread. However, the OIS in CZK is illiquid and thus cannot be used as a good proxy for expectations of future policy rates anyway.

²¹ Liquid assets are defined by the CNB and comprise cash, claims on the CNB, government bonds, and short-term interbank claims.

ratio of non-performing loans, the loan loss provisions ratio, and the 3M default rate on corporate loans) and one profitability indicator (return on assets, ROA). Finally, this group includes the CDS spreads of the parent banks of reference banks quoting PRIBOR as a variable capturing group-wide pressure related to the group-wide liquidity or solvency position, to reflect possible contagion effects from the parent bank to the domestic reference bank.

The second group of variables serves to test H2 on *counterparty risk*. While it is easy for a bank to judge its own liquidity and solvency position, given that all the necessary data are available internally in the bank, it is much more difficult to get information about potential counterparties for interbank transactions. As a proxy for counterparty solvency, the CDS spreads of the counterparty bank are usually used in the literature. However, the CDS of Czech banks are not traded. Instead, the average counterparty's ratio of non-performing loans (NPL) and the average counterparty's return on assets (ROA) are used, since we believe there is close link between a potential counterparty's level of bad loans and profitability on the one hand and its solvency on the other hand. The NPL ratio is published monthly by the CNB as an aggregate for the banking sector, so each reference bank can roughly estimate the average for the rest of the banking sector using its knowledge of its own NPL ratio and the aggregate NPL ratio. Moreover, NPL data for banking groups (such as large, medium-sized, and small banks and foreign bank branches) are available on a quarterly basis. ROA is publicly available on the CNB website at quarterly frequency for the aggregated banking sector and for some large banks also on an individual basis and is usually closely monitored by banking sector analysts. Ideally, one would also like to include some measure of the dispersion of these variables among potential counterparties, but individual data for all reference banks are not publicly available, hence banks would not be able to calculate it.

This group of variables also includes indicators related to the parent banks of potential counterparties. We construct average CDS spreads across all parent banks of potential counterparties as well as the dispersion of CDS spreads, since individual data are publicly available for this variable to capture the source of potential counterparty risk premium that is due to the contagion effect of parent (foreign) banks.²²

The third group includes variables to test H3 on the effect of domestic and foreign *financial markets*. For domestic markets, we use indicators of bond market liquidity constructed by the CNB, long-term (5Y) government bond spreads vis-à-vis 5Y IRS, and exchange rate volatility. For foreign markets, we use the risk premium in the euro interbank market, constructed as the difference between the 3M EURIBOR and the 3M EUR OIS.

The fourth group of variables captures *interbank market activity* to test H4 and includes the ratio of net interbank market exposure to total assets, the connectivity of the reference bank with other reference banks, and the durability of interbank exposures.

The final group of variables allows for testing various aspects of possible *strategic behavior* (H5). Firstly, we include the share of corporate loans linked to PRIBOR in total assets to see whether banks with a higher share of these loans strategically push PRIBOR up. Secondly, to test whether banks take into account the sensitivity of their derivatives exposures to changes in PRIBOR when submitting quotes (i.e., if a bank would earn when PRIBOR goes up, it could strategically push PRIBOR

²² Interviews with reference bank representatives confirmed that when judging counterparty risk, a bank takes into account the CDS of parent banks.

upwards), a good variable to capture this sensitivity is needed. Unfortunately, the CNB reports on interest rate derivatives only at monthly frequency and only includes their real and nominal values and not such a sensitivity parameter. Thus, we constructed an indicator of sensitivity ourselves in a simplified manner – we divided the monthly changes in the real value of the interest rate derivatives by the monthly changes in the 3M PRIBOR. Positive figures of this measure would indicate that the derivatives portfolio increases with the 3M PRIBOR. A significant and positive coefficient of this measure would be indicative of possible strategic behavior to push up PRIBOR in order to earn. If banks were passive holders of such derivatives, this would capture the sensitivity relatively well. Clearly, however, banks use these derivatives in an active manner and thus the indicator of sensitivity (which is, moreover, only at monthly frequency) is probably a very weak proxy, so the results must be interpreted with caution.

5.2 Methodology

With respect to the time and cross-sectional dimension of the data set, our preferred approach to exploring the relevant determinants of the interbank market spread is the panel estimation method. We opt for the fixed-effects model, allowing for unobserved effects specific to the individual reference banks. The reduced-form equation in matrix notation can be expressed as:

spread =
$$c_{\gamma} + X_{\gamma}\beta_{\gamma} + \varepsilon \quad \varepsilon \sim (0, \sigma^2 I)$$
,

where *spread* is the vector of the dependent variable, the 3M PRIBOR–2W repo spread, X represents the matrix of explanatory variables (see Appendix A), β_{γ} is the vector of the coefficient to be estimated, and ε denotes the vector of disturbances. Lower-index γ represents a particular model which includes a particular subset of potentially relevant explanatory variables.

The common approach to removing fixed effects from the data is to use a within transformation. Alternatively, one may include dummy variables for individual reference banks in order to capture bank-specific effects. Since we have up to 13 cross sections (banks) and the number of time periods is high in our sample, we are able to use the dummy variable approach without losing a significant number of degrees of freedom. The estimated equation can be rewritten into:

spread =
$$c_{\gamma} + X_{\gamma}\beta_{\gamma} + D_{\gamma}\eta_{\gamma} + \varepsilon$$
,

where matrix D represents dummies for individual cross sections (banks). Although the fixed-effect technique seems to be appropriate for addressing our task, we still face high uncertainty about the "true" model which reflects the decision-making process of reference banks when quoting PRIBOR and hence influencing the interbank market spread. We identify 23 potentially relevant explanatory variables (excluding dummies). To examine all the potential models, we would have to run 2^{23} regressions, which would be computationally demanding.

Therefore, we take advantage of the Bayesian Model Averaging (BMA) method in combination with the Markov Chain Monte Carlo Model Composition (MC³) algorithm, which can effectively tackle the issue of a large number of potential explanatory variables.²³ This method is becoming popular also in the finance literature for dealing with the problem of model uncertainty (Sousa and Sousa,

²³ For more details on MC³ see Madigan and York (1995). The computations of BMA in combination with MC³ were done using the BMS package by Feldkircher and Zeugner (2009).

2011; Crespo-Cuaresma et al., 2010). The crucial outputs of BMA are the posterior model probability (PMP), the posterior inclusion probability (PIP) of each variable, and the posterior mean (PM) of the coefficients. Posterior model probabilities denote the probability that a particular model is the "true" model. Given the prior model probability $p(M_{\gamma})$, the PMP can be calculated using Bayes' Rule:

$$p(M_{\gamma} | y, X) = \frac{p(y | M_{\gamma}, X) p(M_{\gamma})}{\sum_{s=1}^{2^{\kappa}} p(y | M_{s}, X) p(M_{s})},$$

where $p(y|M_{\gamma}, X)$ is the marginal likelihood of the model and $p(M_{\gamma})$ is the prior model probability reflecting our prior beliefs about the model space. We use the binomial-beta model prior as suggested by Ley and Steel (2009). The denominator (integrated likelihood), given by the sum of the marginal likelihoods of all individual models weighted by the prior model probability, is constant over all models. The marginal likelihood can in theory be calculated in the following way:

$$p(y \mid M_{\gamma}, X) = \int p(y \mid \theta^{\gamma}, M_{\gamma}) p(\theta^{\gamma} \mid M_{\gamma}) d\theta^{\gamma},$$

where $p(y | \theta^{\gamma}, M_{\gamma})$ is the likelihood and $p(\theta^{\gamma} | M_{\gamma})$ represents the prior density of the parameter vector θ^{γ} under model M_{γ} . Setting prior densities is of crucial importance. To reflect the lack of knowledge, we set a non-informative prior on the intercept and error variance, thus $p(c_{\gamma}) \propto 1$ and $p(\sigma) \propto \sigma^{-1}$. Further, we employ Zellner's g-prior (Zellner, 1986) for parameters β_{γ} with a benchmark hyper-prior on g according to Fernández et al. (2001) in the form of $g = \max(N, K^2)$, where N is the number of observations and K is the total number of covariates.

PIP determines the robustness of a variable with respect to the dependent variable and is calculated as the sum of the PMPs of models where the variable was included, formally:

$$PIP = p(\beta_{\gamma} \neq 0 \mid y, X) = \sum_{\beta_{\gamma} \neq 0} p(M_{\gamma} \mid y, X)$$

The PM of a coefficient β is calculated as the weighted average, with the weights being the PMPs, formally:

$$PM = E(\beta \mid y, X) = \sum_{\gamma=1}^{2^{\kappa}} E(\beta_{\gamma} \mid y, X, M_{\gamma}) p(M_{\gamma} \mid y, X)$$

Additionally, we report the standardized version of the posterior mean (SPM) of the coefficients, which might provide more helpful information on the importance of the explanatory variable with respect to the dependent variable.²⁴ Finally, we also report the posterior standard deviation of the coefficients and their conditional posterior sign.

A special approach is applied to address the uncertainty about the appropriate number of lags of market-based variables included in the BMA estimation.²⁵ In principle, it would be possible to

²⁴ Standardized coefficients arise if both the dependent variable and the explanatory variables are normalized (zero mean and variance of one), hence effectively bringing the data down to the same order of magnitude.

²⁵ For other variables, expert-based judgment is used to determine the optimal number of lags.

include, say, up to five lags of all market-based variables. However, this would substantially increase the number of models and possibly introduce multicollinearity into the models, to which BMA is not robust. Hence, we employ the panel VAR (PVAR²⁶) approach, similarly to Babecký et al. (2011), which allows for heterogeneity among individual cross sections. The PVAR approach should suggest how fast a change on a given market is reflected in the domestic interbank risk premium. The estimated bivariate PVAR with 5 lags is of the form:

 $spread_d_{it} = c_1 + \alpha_{1,1}spread_d_{it-1} + ... + \alpha_{1,5}spread_d_{it-5} + \beta_{1,1}market_d_{it-1} + ... + \beta_{1,5}market_d_{it-5} + \varepsilon_{1,it}$ $market_d_{it} = c_2 + \alpha_{2,1}spread_d_{it-1} + ... + \alpha_{2,5}spread_d_{it-5} + \beta_{2,1}market_d_{it-1} + ... + \beta_{2,5}market_d_{it-5} + \varepsilon_{2,it}$

where *spread* d_{it} stands for the change in the risk premium for bank *i* at time *t* and *market* d_{it} is the change in the respective market variable. The optimal lag of the market-based variables is set based on the response of the dependent variable in terms of magnitude and the sign to the shock to the respective market variable. The results of the PVAR analysis are provided in Appendix A (column "Lag(method)") and Appendix B (impulse-response functions).

6. Results

We run the BMA on the period 2007–2012 using daily data, given that the quoting behavior is a process that happens daily, with banks judging daily market developments as well as less frequent balance-sheet information. Each model includes dummy variables representing reference-bank-specific effects which were not subjected to the BMA selection procedure.²⁷ The results indicate strong persistence of the risk premium, as evidenced by the large standardized posterior mean (SPM) of the lagged risk premium. The control variable for expectations of future monetary policy has large posterior inclusion probability (PIP) and the expected sign.

As to the individual hypotheses, the results can be summarized as follows (see Figure 5; a table with detailed underlying BMA results, including PIP, PM, and SPM, can be found in Appendix C): First, we do not find strong support for the funding liquidity risk hypothesis H1. The only variable from this group of explanatory variables that has large PIP is the provisions ratio (*provisionsratio*), with the correct sign, indicating some role of indirect funding liquidity risk through decreased solvency.

Second, two variables on counterparty risk, namely, the average NPL ratio of potential counterparties (*nplratioavg_l*) and the dispersion of the CDS spreads of foreign parent banks of potential counterparties (*cdsstd_l*), have high PIP and the correct signs. This can be considered supportive of H2 on the role of counterparty risk. The latter variable is also indicative of possible cross-border contagion effects via foreign ownership links, causing the Czech interbank market risk premium to increase if the dispersion of the CDS spreads of foreign parent banks increases. This is in line with Heider et al. (2009), who suggest that the dispersion of riskiness among potential counterparties is crucial during turbulent times. The average of the CDS spreads of foreign parent banks of potential

²⁶ The program for PVAR was written and first used by Love and Zicchino (2006).

²⁷ The number of dummy variables representing reference-bank-specific effects corresponds to the number of reference banks operating within a given period minus one in order to avoid perfect multicollinearity, since each model also includes an intercept. In this case, the intercept represents a benchmark bank to which other reference-bank-specific effects are compared.

counterparties (*cdsavg_l*) has large PIP, too, but the incorrect sign. Michaud and Upper (2008) suggest that it takes longer for the information in CDS spreads to feed into the interbank risk premium. Moreover, our results indicate that the dispersion of counterparty riskiness might be more relevant for the interbank risk premium than the average counterparty riskiness within a short period of time, as supported by the higher PIP of *cdsstd_l* compared to *cdsavg_l*. This result is rather intuitive, since in the Czech case the counterparty bank might be any of the reference banks, which are quite heterogeneous in their characteristics (see section 3.2).

Third, market variables – in particular domestic bond market liquidity (bm_l) and the euro area money market risk premium (*euribois3m_l*) – are relevant for the determination of the domestic interbank risk premium (hypothesis H3). Fourth, the connectivity indicators (*connectivity* and *relationship*) have low PIP, thus not supporting H4. Finally, we do not find any strong evidence of strategic behavior in the BMA analysis (hypothesis H5).



Figure 5: Results of BMA Based on Daily Data (1/1/2007-6/30/2012)

Note: This figure shows the 1000 best models indicated by the BMA procedure resulting from 2 million iterations after 1 million burn-ins. The order of the variables is set according to their PIP. The colors in the figure represent the sign of the coefficient, i.e., blue (+), red (-). The dummy variables representing reference-bank-specific effects are included in each model by default, i.e., they are not subjected to the selection procedure.

Source: Authors' calculation

We look additionally at two sub-periods of our sample. Since the pre-crisis period 2007–September 2008 is characterized by low cross-sectional volatility of individual PRIBOR quotes supported by the convention of a 10 bp bid-ask spread, we run the BMA only for the sub-periods after September 2008, namely, September 15, 2008–June 2009 (the crisis period) and July 2009–June 2012 (the recession period) – see Figure 6 and Table 1. The crisis period is the turbulent period following the collapse of Lehman Brothers, characterized by significant uncertainty. The recession period mainly covers the low interest rate environment and economic recession in the Czech Republic and abroad, but with a high level of uncertainty about future economic developments. The results for the two sub-samples can be found in Appendix D.

Figure 6: Individual Quotes of Reference Banks for 3M PRIBOR Less 2W Repo (bp)



Source: Thomson Reuters, authors' calculation

Table 1: Estimation Details

Dependent variable (pp)	3M PRIBOR quote - 2W repo
Period	1/1/2007–6/30/2012
Number of cross sections (reference banks)	13
Number of time periods (daily/monthly)	1435/66
Number of explanatory variables	22*
Crisis subperiod	9/15/2008-6/30/2009
Recession subperiod	7/1/2009–6/30/2012

Note: *Due to its relatively low variation and hence relatively high correlation with reference banks' dummy variables, market share has been excluded from the initial set of variables examined by BMA.

In the *crisis period*, the most important explanatory variable (next to the lagged risk premium and FRA to control for expectations about monetary policy) is bond market liquidity, confirming the impact of the bond market crisis in October 2008 on banks' behavior. In the *recession period*, the counterparty credit risk variables – mainly the dispersion of the CDS spreads of parent banks of potential counterparties and the average ROA of potential counterparties – take over. This is in line with the characteristics of this sub-period. The euro area interbank market risk premium has large PIP, too, but the incorrect sign. The main reason could be that the stress in the euro area banking system reflecting the government debt crisis in 2010–2011 did not affect the Czech banking sector which experienced even slight decrease in the interbank risk premium in that time (see Figure 3).

As a robustness check, we performed the BMA analysis for the whole period on monthly data (see Figure 7; a table with detailed underlying BMA results, including PIP, PM, and SPM, can be found in Appendix C). The main reason for running monthly estimations is to exclude the possibility that the various balance-sheet indicators which are originally at monthly frequency have lower variability visà-vis daily market variables and thus could end up as unimportant in the daily estimations. The monthly results broadly confirm the main results of the analysis performed on daily data, especially for H2 on counterparty risk and H3 on market variables. The negative sign for the coefficient for exchange rate volatility, which was also observed in the daily estimations but with much lower PIP, might reflect a preference for domestic interbank lending relative to interbank lending abroad in an environment of increased exchange rate uncertainty.



Figure 7: Results of BMA Based on Monthly Data (January 2007–June 2012)

Cumulative Model Probabilities

Note: This figure shows the 1000 best models indicated by the BMA procedure resulting from 2 million iterations after 1 million burn-ins. The order of the variables is set according to their PIP. The colors in the figure represent the sign of the coefficient, i.e., blue (+), red (-). The dummy variables representing reference-bank-specific effects are included in each model by default, i.e., they are not subjected to the selection procedure.

Source: Authors' calculation

To fully exclude the possibility of strategic behavior we also examine the cross-sectional variability rather than the within-variation, because part of the cross-sectional variability causing strategic behavior might be hidden in the fixed effects. Therefore, we employ analysis based on the estimated fixed effects (dummies for individual reference banks, *dummy_1-dummy_12*). We divide reference banks into two groups depending on whether their bank-specific intercept (i.e., the sum of the common intercept and the bank-specific dummy) is above or below the average of the bank-specific intercepts.²⁸ For these two groups, we report separately the average ratio of liquid assets to total assets

²⁸ The former group of banks includes banks with an above-average intercept identified in both the daily and monthly estimations.

and the average market share to test for possible predatory behavior, and the average ratio of PRIBOR-linked corporate loans to total assets to test for strategic behavior aiming at increasing revenues from PRIBOR-linked loans.²⁹ Table 2 shows that banks with above-average bank-specific intercepts have on average better liquidity ratios and large market shares, suggesting the possibility of predatory behavior (larger and more liquid banks quoting higher PRIBOR and earning more on lending interbank funds). The average results also indicate that there might have been some strategic behavior in terms of quoting higher PRIBOR rates to increase interest income from PRIBOR-linked loans in loan portfolios.

	Averages over 1/1/2007–6/30/2012			
	Predatory beha	vior	Intentions to increase profits	
	Ratio of liquid assets to total assets (%)	Market share (%)	Ratio of PRIBOR-linked corporate loans to total assets (%)	
Banks with above-average intercept	24.2	7.7	21.3	
Banks with below-average intercept	21.4	4.1	17.4	

Table 2: Analysis Based of	the Estimated	Fixed Effects
----------------------------	---------------	---------------

Source: CNB, authors' calculation

Nevertheless, the results of the analysis based on the bank-specific intercepts must be interpreted with caution. First, the differences among bank-specific intercepts are very low, so that the economic significance of the potential strategic behavior is also very low. Second, a detailed look at the bank-specific intercepts suggests that there is quite some heterogeneity in banks' characteristics within the two groups. Thus, while on average the banks with above-average intercepts have higher liquidity, there are also banks in this group with relatively low liquidity, and the other way round. This also holds for market share and PRIBOR-linked loans, so that these factors are not necessarily the only ones to contribute to above-average intercepts. There might also be other factors playing a role in the cross-sectional variability, often working against strategic behavior, for example possible higher risk aversion of larger and more liquid banks. Thus, our findings provide relatively weak support for strategic behavior. Finally, if only a small number of banks give above-average quotes, the effect on the final PRIBOR would be reduced by dropping the two highest quotes (CNB, 2006), effectively preventing the manipulation of PRIBOR.

7. Conclusion

The global financial crisis has featured an increase in risk premium in interbank markets worldwide. This paper focuses on explaining the risk premium – defined as the departure of the reference 3M PRIBOR from the monetary policy 2W repo rate – in the Czech interbank market. Given the structural features of the Czech banking sector (a comfortable solvency and liquidity situation of banks and foreign ownership as a potential channel for contagion), we tested a number of specific hypotheses on banks' behavior in terms of their quoting contributing rates for PRIBOR, using banklevel, sector-level, and market data. Our results, based on the Bayesian Model Averaging method, can be summarized as follows:

²⁹ The averages are constructed first as averages across time and then as averages across banks with 1) an aboveaverage intercept and 2) a below-average intercept.

First, funding liquidity risk measured by various liquidity and solvency indicators based on balancesheet ratios is not found to influence the interbank risk premium significantly. Although some liquidity tensions might have been felt by the banks, the results could reflect the fact that domestic banks have long had comfortable liquidity positions and thus liquidity risk did not increase sufficiently in the crisis and recession period to be fully reflected in the interbank market risk premium.

Second, the explanatory variables used to capture counterparty risk proved to be important, especially in the period after 2009. Some of these variables are based on the characteristics of foreign parent banks of Czech banks, indicating possible cross-border contagion effects influencing the Czech interbank market premium. Third, market variables reflecting risks abroad and risks in the domestic bond market played an important role in the quoting behavior of reference banks. Fourth, the results do not suggest any high relevance of reference banks' interbank lending relations. Finally, we do not find robust evidence for strategic behavior of banks when quoting PRIBOR rates.

References

- AFONSO, A. AND M. AUBYN (1998): "Credit Rationing and Monetary Transmission: Evidence for Portugal." ISEG Economics Department Working Paper No. 7/98.
- ACHARYA, V. AND O. MERROUCHE (2010): "Precautionary Hoarding of Liquidity and Inter-Bank Markets: Evidence from the Subprime Crisis." NBER WP No. w16395.
- ACHARYA, V., D. GROMB, AND T. YORULMAZER (2010A): "Imperfect Competition in the Interbank Market for Liquidity as a Rationale for Central Banking." NYU Working Paper No. FIN-09-031.
- ACHARYA, V., D. GALE, AND T. YORULMAZER (2010B): "Rollover Risk and Market Freezes." NBER WP No. 15674.
- ANGELINI, P., A. NOBILI, AND C. PICILLO (2009): "The Interbank Market After August 2007: What Has Changed, and Why?" Bank of Italy WP No. 731.
- BABECKÝ, J., T. HAVRÁNEK, J. MATĚJŮ, M. RUSNÁK, K. ŠMÍDKOVÁ, AND B. VAŠÍČEK (2011): "Early Warning Indicators of Economic Crises: Evidence from a Panel of 40 Developed Countries." CNB Working Paper No. 8/2011.
- BÁRTA, V. AND M. SINGER (2006): "The Banking Sector after 15 Years of Restructuring: Czech Experience and Lessons." *BIS Papers* 28, pp. 203–212.
- CABALLERO, R. AND A. SIMSEK (2009): "Complexity and Financial Panics." NBER WP No. 14997.
- CNB (2006): "Rules for Reference Banks and the Calculation (Fixing) of Reference Interest Rates (PRIBID and PRIBOR)."
- CNB (2012): Financial Stability Report 20011/2012.
- CRESPO-CUARESMA, J., A. GERŠL, AND T. SLAČÍK (2010): "Could Markets Have Helped Predict the Puzzling Exchange Rate Path in CESEE Countries During the Current Crisis?" *OeNB Focus on European Economic Integration* Q1/10, pp. 32–48.
- FELDKIRCHER, M. AND S. ZEUGNER (2009): "Benchmark Priors Revisited: On Adaptive Shrinkage and the Supermodel Effect in Bayesian Model Averaging." IMF Working Paper 09-202.
- FRAIT, J., A. GERŠL AND J. SEIDLER (2011): "Credit Growth and Financial Stability in the Czech Republic." World Bank Policy Research Working Paper No 5771.
- FREIXAS, X. AND J. JORGE (2008): "The Role of Interbank Markets in Monetary Policy: A Model with Rationing." *Journal of Money, Credit and Banking* 40(6), pp. 1151–1176.
- GAI, P., A. HALDANE, AND S. KAPADIA (2011): "Complexity, Concentration and Contagion." *Journal of Monetary Economics* 58(5), pp. 453–470.

- GERŠL, A. AND Z. KOMÁRKOVÁ (2009): "Liquidity Risk and Banks' Bidding Behavior: Evidence from the Global Financial Crisis." *Czech Journal of Economics and Finance* 59(6), pp. 577–592.
- HEIDER, F., M. HOEROVA, AND C. HOLTHAUSEN (2009): "Liquidity Hoarding and Interbank Market Spreads: The Role of Counterparty Risk?" EBC Discussion Paper No. 2009-11S.
- KAMPS, A. (2009): "What Drives International Money Market Rates? Lessons from a Cointegration VAR Approach." ECB, forthcoming.
- LEY, E. AND M. STEEL (2009): "On the Effect of Prior Assumptions in Bayesian Model Averaging with Applications to Growth Regression." *Journal of Applied Econometrics* 24(4), pp. 651–674.
- LOVE, I. AND L. ZICCHINO (2006): "Financial Development and Dynamic Investment Behavior: Evidence from Panel VAR." *The Quarterly Review of Economics and Finance* 46(2), pp. 190–210.
- MADIGAN, D. AND J. YORK (1995): "Bayesian Graphical Models for Discrete Data." *International Statistical Review* 63(2), pp. 215–232.
- MICHAUD, F. AND C. UPPER (2008): "What Drives Interbank Rates? Evidence from the Libor Panel." BIS Quarterly Review.
- NOBILI, S. (2009): "Liquidity Risk in Money Market Spreads." mimeo.
- SOUSA, J. AND R. M. SOUSA (2011): "Asset Returns Under Model Uncertainty: Evidence From the Euro Area, the U.K. and the U.S." Banco de Portugal Working Paper No 19.
- TABOGA, M. (2013): "What is a Prime Bank? A Euribor–OIS Perspective." Banka d'Italia Working Paper No. 895.
- TAYLOR, J. AND J. WILLIAMS (2008): "A Black Swan in the Money Market?" NBER WP No. 13943.
- ZELLNER, A. (1986): "On Assessing Prior Distributions and Bayesian Regression Analysis with Gprior Distributions." *Bayesian Inference and Decision Techniques: Essays in Honor of Bruno de Finetti* (eds. P. K. Goel and A. Zellner), pp. 233–243.

				Expected		Linearly	Original
Category	Shortcut	Variable	Description	Sign	Lag (method)	interpolated	frequency
Dependent	spread	3M PRIBOR quote less 2W REPO (bp)					daily
Lag of dependent variable	spread_I	One-day lag of 3M PRIBOR quote less 2W REPO (bp)					daily
policy expectations	fra1v4	Forward rate agreement (FRA 1x4, %)			3 days (VAR)		daily
	liquidratio	Ratio of liquid assets to total assets (%)	Liquidity risk of a	-		LI	monthly
	Itdratio	Loan-to-deposit ratio (%)	reference bank	+		LI	monthly
	roa	Return on assets (%)				LI	monthly
(1) Funding liquidity risk	npl	Ratio of NPLs to total client loans (%)	Ore ditainly of a	+		LI	monthly
	provisionsratio	Ratio of loan loss provisions to total client loans (%)	credit risk of a	+		LI	monthly
	defaultnpl3m	3M default rate of corporate loans (%)	Telefence bank	+		LI	monthly
	cdsbank_I	5Y CDS of parent bank (bp)		+	4 days (VAR)		daily
	roaratioavg_l	Average return on assets across counterparty reference banks (%)		-	1 month (expert-based)		quarterly
	nplratioavg_l	Average ratio of NPLs to total client loans across counterparty reference banks (%)	Credit risk of	+	1 month (expert-based)		monthly
(2) Counterparty risk	cdsavg_l	Average 5Y CDS spreads across parent banks of counterparty reference banks (bp)	counterparty	+	1 day (VAR)		daily
	cdsstd_I	Standard deviation of 5Y CDS spreads across parent banks of counterparty reference banks divided by mean (%)		+	1 day (VAR)		daily
	bm_l	Bond market liquidity index		-	2 days (VAR)		daily
(3) Financial markets	gb5y_l	Risk premium on 5Y bonds (5Ygovernment bond yield less 5Y IRS, bp)	Domestic markets	+	3 days (VAR)		daily
	exvol3m_l	3M implied exchange rate volatility CZK/EUR (%)		_ +	4 days (VAR)		daily
	euribois3m_l	Risk premium on euro interbank market (3M EURIBOR less 3M EUR OIS, bp)	Foreign market	+	2 days (VAR)		daily
	netinterbankratio	Net interbank exposure to total assets (%)		-		LI	monthly
(4) Interbank market activity	connectivity	Connectivity of a reference bank with other reference banks (%)		-		LI	quarterly
、, ,	relationship⁺	reference banks (%)		-		LI	quarterly
	floatingtoassets	Ratio of PRIBOR-linked corporate loans to total assets (%)		+		LI	monthly
(5) Strategic behavior	derivatsensm	Sensitivity of derivatives to change in PRIBOR index (thousands of CZK)		+		LI	monthly
	marketshare	Market share of a bank according to assets (%)	<u> </u>	+		LI	monthly
Bank dummies	dummy_x	Dummies for individual reference banks	Bank-specific				daily

Appendix A: List of Explanatory Variables

Note: This table is related to the analysis based on daily data. For the analysis based on monthly data, monthly averages of daily variables were calculated without assuming any lag of these variables except for a one-month lag of the dependent variable.

- The bond market liquidity index is normalized (i.e., expressed in standard deviations from the historical mean) and moves roughly between 2.5 and -2.5 with zero mean and a standard deviation of one.
- The variables expressed in percent are to be understood in units of percent (for example, a value of 26.2 for 26.2%).
- Connectivity is calculated as the ratio of the number of lending relations with other reference banks to the maximum number of relations which can potentially be created with other reference banks. ⁺Relationship lending is defined as the ratio of lending relations with other reference banks lasting for one year to total lending relations with other reference banks existing in a given period.



Appendix B: Results of PVAR Used for the Determination of the Optimal Lag of Market-based Variables

Note: Each figure represents the impulse response functions of the dependent variable (spread_d) resulting from the PVAR with five lags to a shock to a particular market variable (the market variables are mentioned at the top of each figure). The impulse response functions of the market variables to spread_d are not reported in order to limit the number of output figures.

Appendix C: Underlying Results of BMA

Results of BMA based on daily data (1/1/2007-6/30/2012)					Results of BM	A bas	ed on mor	nthly data (January	2007-June 201	2)	
		Posterior	Posterior standard	Posterior mean	Conditional			Posterior	Posterior standard	Posterior mean	Conditional
	PIP	mean	deviation	(stan dardized)	posterior sign		PIP	mean	deviation	(stan dardized)	posterior sign
spread_I	1.00	0.9360	0.0026	0.9369	1.00	spread_I	1.00	0.6093	0.0192	0.6232	1.00
bm_l	1.00	-1.0804	0.1095	-0.0281	0.00	fra1x4	1.00	7.4256	1.7795	0.3581	1.00
euribois3m_l	1.00	0.0171	0.0028	0.0248	1.00	bm	1.00	-23.6528	1.8384	-0.5945	0.00
nplratioavg_l	1.00	0.7565	0.0796	0.0438	1.00	exvol3m	1.00	-2.1421	0.2800	-0.2962	0.00
dummy_1	1.00	0.1173	0.2116	0.0013	0.86	nplratioavg_l	1.00	5.7582	0.7261	0.3522	1.00
dummy_2	1.00	0.2850	0.2281	0.0033	1.00	dummy_1	1.00	1.6234	1.5283	0.0197	1.00
dummy_3	1.00	-0.1692	0.1966	-0.0019	0.02	dummy_2	1.00	2.9081	1.8160	0.0353	1.00
dummy_4	1.00	0.0391	0.2159	0.0004	0.50	dummy_3	1.00	-0.1462	1.5255	-0.0018	0.20
dummy_5	1.00	0.3411	0.1787	0.0039	1.00	dummy_4	1.00	0.9235	1.5874	0.0112	0.87
dummy_6	1.00	-0.0935	0.2486	-0.0011	0.27	dummy_5	1.00	2.1935	1.4417	0.0267	1.00
dummy_7	1.00	0.1533	0.2574	0.0018	0.82	dummy_6	1.00	-1.2200	2.0670	-0.0148	0.14
dummy_8	1.00	0.3428	0.1875	0.0039	1.00	dummy_7	1.00	2.3072	1.6445	0.0280	1.00
dummy_9	1.00	0.1678	0.2126	0.0019	0.95	dummy_8	1.00	2.0711	1.6116	0.0252	1.00
dummy_10	1.00	-0.0037	0.2244	0.0000	0.07	dummy_9	1.00	1.4049	1.7973	0.0167	0.86
dummy_11	1.00	0.1028	0.2450	0.0007	0.99	dummy_10	1.00	0.4339	2.2181	0.0053	0.22
dummy_12	1.00	-0.0768	0.4821	-0.0002	0.04	dummy_11	1.00	0.3243	1.9655	0.0025	0.94
fra1x4_I	1.00	0.7165	0.1525	0.0328	1.00	dummy_12	1.00	0.3007	3.8516	0.0010	0.76
cdsstd_I	1.00	0.0210	0.0044	0.0109	1.00	euribois3m	0.61	0.0456	0.0429	0.0688	1.00
cdsavg_l	0.73	-0.0027	0.0019	-0.0092	0.00	cdsavg	0.42	-0.0137	0.0176	-0.0485	0.00
provisionsratio	0.53	0.0773	0.0792	0.0053	1.00	gb5y	0.38	0.0305	0.0429	0.0432	1.00
gb5y_l	0.30	0.0023	0.0038	0.0032	1.00	provisionsratio	0.16	0.1200	0.3047	0.0088	1.00
cdsbank_l	0.16	-0.0003	0.0007	-0.0012	0.00	floatingtoassets	0.14	0.0333	0.0905	0.0125	1.00
exvol3m_l	0.08	-0.0059	0.0212	-0.0008	0.00	relationship	0.05	-0.0011	0.0061	-0.0012	0.00
floatingtoassets	0.05	0.0016	0.0074	0.0006	1.00	liquidratio	0.04	0.0021	0.0133	0.0010	1.00
derivatsensm	0.02	0.0000	0.0000	-0.0001	0.00	cdsbank	0.03	-0.0002	0.0016	-0.0009	0.01
nplratio	0.02	0.0007	0.0054	0.0001	1.00	nplratio	0.03	0.0044	0.0355	0.0006	0.96
relationship	0.01	0.0000	0.0004	0.0000	0.00	roaratioavg_l	0.03	0.1082	1.0128	0.0005	0.90
roa	0.01	-0.0010	0.0129	0.0000	0.00	netinterbankratio	0.02	0.0019	0.0177	0.0004	1.00
connectivity	0.01	0.0000	0.0003	0.0000	0.92	cdsstd	0.02	0.0008	0.0085	0.0004	0.83
netinterbankratio	0.01	0.0001	0.0013	0.0000	1.00	roa	0.02	-0.0076	0.1001	-0.0002	0.01
Itdratio	0.01	0.0000	0.0002	0.0000	0.98	defaultnpl3m	0.02	-0.0030	0.0439	-0.0002	0.00
roaratioavg_l	0.01	-0.0006	0.0429	0.0000	0.40	connectivity	0.02	0.0002	0.0032	0.0002	0.89
liquidratio	0.01	0.0000	0.0004	0.0000	0.77	Itdratio	0.02	0.0000	0.0022	0.0000	0.74
defaultnpl3m	0.00	0.0000	0.0026	0.0000	0.85	derivatsensm	0.02	0.0000	0.0000	0.0000	0.54
intercept	1.00	-3.2084		-0.1305		intercept	1.00	-11.7948		-0.5064	

Appendix D: Results of BMA for the Crisis and Recession Periods (Daily Data)



Results of BMA for the crisis period (9/15/2008-6/30/2009)

Results of BMA for the recession period (7/1/2009–6/30/2012)



Cumulative Model Probabilities

Cumulative Model Probabilities

Note: This figure shows the 1000 best models indicated by the BMA procedure resulting from 2 million iterations after 1 million burn-ins. The order of the variables is set according to their PIP. The colors in the figure represent the sign of the coefficient, i.e., blue (+), red (-). The dummy variables representing reference-bank-specific effects are included in each model by default, i.e., they are not subjected to the selection procedure.
Source: Authors' calculation

CNB WORKING PAPER SERIES

1/2013	Adam Geršl Jitka Lešanovská	Explaining the Czech interbank market risk premium
15/2012	Róbert Ambriško Jan Babecký Jakub Ryšánek Vilém Valenta	Assessing the impact of fiscal measures on the Czech economy
14/2012	Václav Hausenblas Ivana Kubicová Jitka Lešanovská	Contagion risk in the Czech financial system: A network analysis and simulation approach
13/2012	Michal Franta	Macroeconomic effects of fiscal policy in the Czech Republic: Evidence based on various identification approaches in a VAR framework
12/2012	Konstantin Belyaev Aelita Belyaeva Tomáš Konečný Jakub Seidler Martin Vojtek	Macroeconomic factors as drivers of LGD prediction: Empirical evidence from the Czech Republic
11/2012	Adam Geršl Petr Jakubík Tomáš Konečný Jakub Seidler	Dynamic stress testing: The framework for testing banking sector resilience used by the Czech National Bank
10/2012	Tomáš Havránek Marek Rusnák	Transmission lags of monetary policy: A meta-analysis
9/2012	Volha Audzei František Brázdik	Monetary policy and exchange rate dynamics: The exchange rate as a shock absorber
8/2012	Alexis Derviz Jakub Seidler	Coordination incentives in cross-border macroprudential regulation
7/2012	Peter Claeys Bořek Vašíček	Measuring sovereign bond spillover in Europe and the impact of rating news
6/2012	Michal Franta Jan Libich Petr Stehlík	Tracking monetary-fiscal interactions across time and space
5/2012	Roman Horváth Jakub Seidler Laurent Weill	Bank capital and liquidity creation: Granger causality evidence
4/2012	Jaromír Baxa Miroslav Plašil Bořek Vašíček	Changes in inflation dynamics under inflation targeting? Evidence from Central European countries
3/2012	Soňa Benecká Tomáš Holub Narcisa Liliana Kadlčáková Ivana Kubicová	Does central bank financial strength matter for inflation? An empirical analysis
2/2012	Adam Geršl Petr Jakubík Dorota Kowalczyk Steven Ongena José-Luis Peydró Alcalde	Monetary conditions and banks' behaviour in the Czech Republic

1/2012	Jan Babecký Kamil Dybczak	<i>Real wage flexibility in the European Union: New evidence from the labour cost data</i>
15/2011	Jan Babecký Kamil Galuščák Lubomír Lízal	Firm-level labour demand: Adjustment in good times and during the crisis
14/2011	Vlastimil Čadek Helena Rottová Branislav Saxa	Hedging behaviour of Czech exporting firms
13/2011	Michal Franta Roman Horváth Marek Rusnák	Evaluating changes in the monetary transmission mechanism in the Czech Republic
12/2011	Jakub Ryšánek Jaromír Tonner Osvald Vašíček	Monetary policy implications of financial frictions in the Czech Republic
11/2011	Zlatuše Komárková Adam Geršl Luboš Komárek	Models for stress testing Czech banks ´ liquidity risk
10/2011	Michal Franta Jozef Baruník Roman Horváth Kateřina Šmídková	Are Bayesian fan charts useful for central banks? Uncertainty, forecasting, and financial stability stress tests
9/2011	Kamil Galuščák Lubomír Lízal	The impact of capital measurement error correction on firm-level production function estimation
8/2011	Jan Babecký Tomáš Havránek Jakub Matějů Marek Rusnák Kateřina Šmídková Bořek Vašíček	Early warning indicators of economic crises: Evidence from a panel of 40 developed countries
7/2011	Tomáš Havránek Zuzana Iršová	Determinants of horizontal spillovers from FDI: Evidence from a large meta-analysis
6/2011	Roman Horváth Jakub Matějů	How are inflation targets set?
5/2011	Bořek Vašíček	Is monetary policy in the new EU member states asymmetric?
4/2011	Alexis Derviz	Financial frictions, bubbles, and macroprudential policies
3/2011	Jaromír Baxa Roman Horváth Bořek Vašíček	Time-varying monetary-policy rules and financial stress: Does financial instability matter for monetary policy?
2/2011	Marek Rusnák Tomáš Havránek Roman Horváth	How to solve the price puzzle? A meta-analysis
1/2011	Jan Babecký Aleš Bulíř Kateřina Šmídková	Sustainable real exchange rates in the new EU member states: What did the Great Recession change?
15/2010	Ke Pang Pierre L. Siklos	Financial frictions and credit spreads
14/2010	Filip Novotný Marie Raková	Assessment of consensus forecasts accuracy: The Czech National Bank perspective
13/2010	Jan Filáček	Central bank forecasts as a coordination device

	Branislav Saxa	
12/2010	Kateřina Arnoštová David Havrlant Luboš Růžička Peter Tóth	Short-term forecasting of Czech quarterly GDP using monthly indicators
11/2010	Roman Horváth Kateřina Šmídková Jan Zápal	Central banks' voting records and future policy
10/2010	Alena Bičáková Zuzana Prelcová Renata Pašaličová	Who borrows and who may not repay?
9/2010	Luboš Komárek Jan Babecký Zlatuše Komárková	Financial integration at times of financial instability
8/2010	Kamil Dybczak Peter Tóth David Voňka	Effects of price shocks to consumer demand. Estimating the QUAIDS demand system on Czech Household Budget Survey data
7/2010	Jan Babecký Philip Du Caju Theodora Kosma Martina Lawless Julián Messina Tairi Rõõm	The margins of labour cost adjustment: Survey evidence from European firms
6/2010	Tomáš Havránek Roman Horváth Jakub Matějů	Do financial variables help predict macroeconomic environment? The case of the Czech Republic
5/2010	Roman Horváth Luboš Komárek Filip Rozsypal	Does money help predict inflation? An empirical assessment for Central Europe
4/2010	Oxana Babecká Kucharčuková Jan Babecký Martin Raiser	A gravity approach to modelling international trade in South- Eastern Europe and the Commonwealth of Independent States: The role of geography, policy and institutions
3/2010	Tomáš Havránek Zuzana Iršová	Which foreigners are worth wooing? A Meta-analysis of vertical spillovers from FDI
2/2010	Jaromír Baxa Roman Horváth Bořek Vašíček	How does monetary policy change? Evidence on inflation targeting countries
1/2010	Adam Geršl Petr Jakubík	Relationship lending in the Czech Republic
15/2009	David N. DeJong Roman Liesenfeld Guilherme V. Moura Jean-Francois Richard Hariharan Dharmarajan	Efficient likelihood evaluation of state-space representations
14/2009	Charles W. Calomiris	Banking crises and the rules of the game
13/2009	Jakub Seidler Petr Jakubík	The Merton approach to estimating loss given default: Application to the Czech Republic
12/2009	Michal Hlaváček Luboš Komárek	Housing price bubbles and their determinants in the Czech Republic and its regions

11/2009	Kamil Dybczak Kamil Galuščák	Changes in the Czech wage structure: Does immigration matter?
10/2009	Jiří Böhm Petr Král Branislav Saxa	Percepion is always right: The CNB's monetary policy in the media
9/2009	Alexis Derviz Marie Raková	Funding costs and loan pricing by multinational bank affiliates
8/2009	Roman Horváth Anca Maria Podpiera	Heterogeneity in bank pricing policies: The Czech evidence
7/2009	David Kocourek Filip Pertold	The impact of early retirement incentives on labour market participation: Evidence from a parametric change in the Czech Republic
6/2009	Nauro F. Campos Roman Horváth	Reform redux: Measurement, determinants and reversals
5/2009	Kamil Galuščák Mary Keeney Daphne Nicolitsas Frank Smets Pawel Strzelecki Matija Vodopivec	The determination of wages of newly hired employees: Survey evidence on internal versus external factors
4/2009	Jan Babecký Philip Du Caju Theodora Kosma Martina Lawless Julián Messina Tairi Rõõm	Downward nominal and real wage rigidity: Survey evidence from European firms
3/2009	Jiri Podpiera Laurent Weill	Measuring excessive risk-taking in banking
2/2009	Michal Andrle Tibor Hlédik Ondra Kameník Jan Vlček	Implementing the new structural model of the Czech National Bank
1/2009	Kamil Dybczak Jan Babecký	The impact of population ageing on the Czech economy
14/2008	Gabriel Fagan Vitor Gaspar	Macroeconomic adjustment to monetary union
13/2008	Giuseppe Bertola Anna Lo Prete	<i>Openness, financial markets, and policies: Cross-country and dynamic patterns</i>
12/2008	Jan Babecký Kamil Dybczak Kamil Galuščák	Survey on wage and price formation of Czech firms
11/2008	Dana Hájková	The measurement of capital services in the Czech Republic
10/2008	Michal Franta	<i>Time aggregation bias in discrete time models of aggregate duration data</i>
9/2008	Petr Jakubík Christian Schmieder	Stress testing credit risk: Is the Czech Republic different from Germany?
8/2008	Sofia Bauducco Aleš Bulíř Martin Čihák	Monetary policy rules with financial instability

7/2008	Jan Brůha Jiří Podpiera	The origins of global imbalances
6/2008	Jiří Podpiera Marie Raková	The price effects of an emerging retail market
5/2008	Kamil Dybczak David Voňka Nico van der Windt	The effect of oil price shocks on the Czech economy
4/2008	Magdalena M. Borys Roman Horváth	The effects of monetary policy in the Czech Republic: An empirical study
3/2008	Martin Cincibuch Tomáš Holub Jaromír Hurník	Central bank losses and economic convergence
2/2008	Jiří Podpiera	Policy rate decisions and unbiased parameter estimation in conventionally estimated monetary policy rules
1/2008	Balázs Égert Doubravko Mihaljek	Determinants of house prices in Central and Eastern Europe
17/2007	Pedro Portugal	U.S. unemployment duration: Has long become longer or short become shorter?
16/2007	Yuliya Rychalovská	Welfare-based optimal monetary policy in a two-sector small open economy
15/2007	Juraj Antal František Brázdik	The effects of anticipated future change in the monetary policy regime
14/2007	Aleš Bulíř Kateřina Šmídková Viktor Kotlán David Navrátil	Inflation targeting and communication: Should the public read inflation reports or tea leaves?
13/2007	Martin Cinncibuch Martina Horníková	<i>Measuring the financial markets' perception of EMU enlargement:</i> <i>The role of ambiguity aversion</i>
12/2007	Oxana Babetskaia- Kukharchuk	Transmission of exchange rate shocks into domestic inflation: The case of the Czech Republic
11/2007	Jan Filáček	Why and how to assess inflation target fulfilment
10/2007	Michal Franta Branislav Saxa Kateřina Šmídková	Inflation persistence in new EU member states: Is it different than in the Euro area members?
9/2007	Kamil Galuščák Jan Pavel	Unemployment and inactivity traps in the Czech Republic: Incentive effects of policies
8/2007	Adam Geršl Ieva Rubene Tina Zumer	Foreign direct investment and productivity spillovers: Updated evidence from Central and Eastern Europe
7/2007	Ian Babetskii Luboš Komárek Zlatuše Komárková	Financial integration of stock markets among new EU member states and the euro area
6/2007	Anca Pruteanu-Podpiera Laurent Weill Franziska Schobert	Market power and efficiency in the Czech banking sector
5/2007	Jiří Podpiera Laurent Weill	Bad luck or bad management? Emerging banking market experience
4/2007	Roman Horváth	The time-varying policy neutral rate in real time: A predictor for

		future inflation?
3/2007	Jan Brůha Jiří Podpiera Stanislav Polák	The convergence of a transition economy: The case of the Czech Republic
2/2007	Ian Babetskii Nauro F. Campos	Does reform work? An econometric examination of the reform-growth puzzle
1/2007	Ian Babetskii Fabrizio Coricelli Roman Horváth	Measuring and explaining inflation persistence: Disaggregate evidence on the Czech Republic
13/2006	Frederic S. Mishkin Klaus Schmidt- Hebbel	Does inflation targeting make a difference?
12/2006	Richard Disney Sarah Bridges John Gathergood	Housing wealth and household indebtedness: Is there a household 'financial accelerator'?
11/2006	Michel Juillard Ondřej Kameník Michael Kumhof Douglas Laxton	Measures of potential output from an estimated DSGE model of the United States
10/2006	Jiří Podpiera Marie Raková	Degree of competition and export-production relative prices when the exchange rate changes: Evidence from a panel of Czech exporting companies
9/2006	Alexis Derviz Jiří Podpiera	Cross-border lending contagion in multinational banks
8/2006	Aleš Bulíř Jaromír Hurník	The Maastricht inflation criterion: "Saints" and "Sinners"
7/2006	Alena Bičáková Jiří Slačálek Michal Slavík	Fiscal implications of personal tax adjustments in the Czech Republic
6/2006	Martin Fukač Adrian Pagan	Issues in adopting DSGE models for use in the policy process
5/2006	Martin Fukač	New Keynesian model dynamics under heterogeneous expectations and adaptive learning
4/2006	Kamil Dybczak Vladislav Flek Dana Hájková Jaromír Hurník	Supply-side performance and structure in the Czech Republic (1995–2005)
3/2006	Aleš Krejdl	Fiscal sustainability – definition, indicators and assessment of Czech public finance sustainability
2/2006	Kamil Dybczak	Generational accounts in the Czech Republic
1/2006	Ian Babetskii	Aggregate wage flexibility in selected new EU member states
14/2005	Stephen G. Cecchetti	The brave new world of central banking: The policy challenges posed by asset price booms and busts
13/2005	Robert F. Engle Jose Gonzalo Rangel	The spline GARCH model for unconditional volatility and its global macroeconomic causes
12/2005	Jaromír Beneš Tibor Hlédik Michael Kumhof	An economy in transition and DSGE: What the Czech national bank's new projection model needs

	David Vávra	
11/2005	Marek Hlaváček Michael Koňák Josef Čada	The application of structured feedforward neural networks to the modelling of daily series of currency in circulation
10/2005	Ondřej Kameník	Solving SDGE models: A new algorithm for the sylvester equation
9/2005	Roman Šustek	Plant-level nonconvexities and the monetary transmission mechanism
8/2005	Roman Horváth	Exchange rate variability, pressures and optimum currency area criteria: Implications for the central and eastern european countries
7/2005	Balázs Égert Luboš Komárek	Foreign exchange interventions and interest rate policy in the Czech Republic: Hand in glove?
6/2005	Anca Podpiera Jiří Podpiera	Deteriorating cost efficiency in commercial banks signals an increasing risk of failure
5/2005	Luboš Komárek Martin Melecký	The behavioural equilibrium exchange rate of the Czech koruna
4/2005	Kateřina Arnoštová Jaromír Hurník	The monetary transmission mechanism in the Czech Republic (evidence from VAR analysis)
3/2005	Vladimír Benáček Jiří Podpiera Ladislav Prokon	Determining factors of Czech foreign trade: A cross-section time series perspective
2/2005	Kamil Galuščák Daniel Münich	Structural and cyclical unemployment: What can we derive from the matching function?
1/2005	Ivan Babouček Martin Jančar	<i>Effects of macroeconomic shocks to the quality of the aggregate loan portfolio</i>
10/2004	Aleš Bulíř Kateřina Šmídková	<i>Exchange rates in the new EU accession countries: What have we learned from the forerunners</i>
9/2004	Martin Cincibuch Jiří Podpiera	Beyond Balassa-Samuelson: Real appreciation in tradables in transition countries
8/2004	Jaromír Beneš David Vávra	Eigenvalue decomposition of time series with application to the Czech business cycle
7/2004	Vladislav Flek, ed.	Anatomy of the Czech labour market: From over-employment to under-employment in ten years?
6/2004	Narcisa Kadlčáková Joerg Keplinger	Credit risk and bank lending in the Czech Republic
5/2004	Petr Král	Identification and measurement of relationships concerning inflow of FDI: The case of the Czech Republic
4/2004	Jiří Podpiera	Consumers, consumer prices and the Czech business cycle identification
3/2004	Anca Pruteanu	The role of banks in the Czech monetary policy transmission mechanism
2/2004	Ian Babetskii	EU enlargement and endogeneity of some OCA criteria: Evidence from the CEECs
1/2004	Alexis Derviz Jiří Podpiera	<i>Predicting bank CAMELS and S&P ratings: The case of the Czech Republic</i>

CNB RESEARCH AND POLICY NOTES			
3/2012	Jan Frait Zlatuše Komárková	Macroprudential policy and its instruments in a small EU economy	
2/2012	Zlatuše Komárková Marcela Gronychová	Models for stress testing in the insurance sector	
1/2012	Róbert Ambriško Vítězslav Augusta Dana Hájková Petr Král Pavla Netušilová Milan Říkovský Pavel Soukup	Fiscal discretion in the Czech Republic in 2001-2011: Has it been stabilizing?	
3/2011	František Brázdik Michal Hlaváček Aleš Maršál	Survey of research on financial sector modeling within DSGE models: What central banks can learn from it	
2/2011	Adam Geršl Jakub Seidler	Credit growth and capital buffers: Empirical evidence from Central and Eastern European countries	
1/2011	Jiří Böhm Jan Filáček Ivana Kubicová Romana Zamazalová	<i>Price-level targeting – A real alternative to inflation targeting?</i>	
1/2008	Nicos Christodoulakis	<i>Ten years of EMU: Convergence, divergence and new policy priorities</i>	
2/2007	Carl E. Walsh	Inflation targeting and the role of real objectives	
1/2007	Vojtěch Benda Luboš Růžička	Short-term forecasting methods based on the LEI approach: The case of the Czech Republic	
2/2006	Garry J. Schinasi	Private finance and public policy	
1/2006	Ondřej Schneider	The EU budget dispute – A blessing in disguise?	
5/2005	Jan Stráský	<i>Optimal forward-looking policy rules in the quarterly projection model of the Czech National Bank</i>	
4/2005	Vít Bárta	Fulfilment of the Maastricht inflation criterion by the Czech Republic: Potential costs and policy options	
3/2005	Helena Sůvová Eva Kozelková David Zeman Jaroslava Bauerová	Eligibility of external credit assessment institutions	
2/2005	Martin Čihák Jaroslav Heřmánek	Stress testing the Czech banking system: Where are we? Where are we going?	
1/2005	David Navrátil Viktor Kotlán	The CNB's policy decisions – Are they priced in by the markets?	
4/2004	Aleš Bulíř	External and fiscal sustainability of the Czech economy: A quick look through the IMF's night-vision goggles	
3/2004	Martin Čihák	Designing stress tests for the Czech banking system	
2/2004	Martin Čihák	Stress testing: A review of key concepts	

CNB ECONOMIC RESEARCH BULLETIN

November 2012	Financial stability and monetary policy
April 2012	Macroeconomic forecasting: Methods, accuracy and coordination
November 2011	Macro-financial linkages: Theory and applications
April 2011	Monetary policy analysis in a central bank
November 2010	Wage adjustment in Europe
May 2010	Ten years of economic research in the CNB
November 2009	Financial and global stability issues
May 2009	Evaluation of the fulfilment of the CNB's inflation targets 1998–2007
December 2008	Inflation targeting and DSGE models
April 2008	Ten years of inflation targeting
December 2007	Fiscal policy and its sustainability
August 2007	Financial stability in a transforming economy
November 2006	ERM II and euro adoption
August 2006	Research priorities and central banks
November 2005	Financial stability
May 2005	Potential output
October 2004	Fiscal issues
May 2004	Inflation targeting
December 2003	Equilibrium exchange rate

Czech National Bank Economic Research Department Na Příkopě 28, 115 03 Praha 1 Czech Republic phone: +420 2 244 12 321 fax: +420 2 244 14 278 http://www.cnb.cz e-mail: research@cnb.cz ISSN 1803-7070